Author's response to reviews

Title: The association between psychiatric diagnosis and violent re-offending in adult offenders in the community

Authors:

Martin Grann (c.m.g@telia.com)
John Danesh (john.danesh@dphpc.cam.ac.uk)
Seena Fazel (seena.fazel@psych.ox.ac.uk)

Version: 4 Date: 14 November 2008

Author's response to reviews:

Thank you for clarifying further formatting changes required, which we have now done.

Specifically, you requested (SEE MY RESPONSES IN BOLD):

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. DONE

Email address- please include email addresses for all the authors on the title page. DONE

Corresponding author - One or more corresponding author(s) must be indicated on the title page, and their email address must be given. DONE

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT, Figure legends not Figure Legends. DONE

Tables - HAVE BEEN CREATED USING THE TABLES TOOL IN MS WORD.